Author's response to reviews

Title: Determinants of prenatal anemia in Ethiopia, 2016

Authors:
Abera Abay (abera.abay159@gmail.com)
Haile Yalew (yalewhaile28@gmail.com)
Amare Tariku (amaretariku15@yahoo.com)
Ejigu Gebeye (ejigugebeye@gmail.com)

Version: 3 Date: 02 Jul 2017

Author's response to reviewers

Title: Determinants of prenatal anemia in Ethiopia, 2016

Authors:
Abera Abay (AA): abera.abay159@gmail.com
Haile Woldie Yalew (HWY): yalewhaile28@gmail.com
Amare Tariku (AT): amaretariku15@yahoo.com
Ejigu Gebeye (EG): ejigugebeye@gmail.com

Version: 4 Date: 2 July, 2017

With regards!

From: Amare Tariku
Correspondence: amaretariku15@yahoo.com

Author's response to reviews: see over

To: Archives of Public Health, editorial board

Subject: Submitting a revised version of the manuscript:

Manuscript ID: AOPH-D-16-00132R1

Title: Determinants of prenatal anemia in Ethiopia, 2016

POINT BY POINT RESPONSES

We would like to take this opportunity to thank the reviewers and editors for their view and forwarding constructive comments. The comments are very important which will further improve the quality of the manuscript. Based on the comments the point–by–point responses are given below.

Reviewer#3

Reviewers comment Response to comments

1. Please delete "Asosa Zone" from the title Answer: Thank you dear reviewer. I corrected the title as per the comment. Also the year of study was added in the title

Editor reviewers comment Response to comments

1. Change title as suggested by reviewer but also add the information of the period the study was done. Answer: Thank you dear editor. The title is modified based on the comment and study period is added.

Thank you!!!!